Author's response to reviews

Title: Costs of the 'Hartslag Limburg' community heart health intervention

Authors:

Emma T Ronckers (s.ronckers@beoz.unimaas.nl)
Wim Groot (W.groot@beoz.unimaas.nl)
Mieke Steenbakkers (m.steenbakkers@ggd-zzl.nl)
Erik Ruland (ERuland@nigz.nl)
Andre Ament (A.Ament@beoz.unimaas.nl)

Version: 7 Date: 16 February 2006

Author's response to reviews:

The title and abstract in the submission system are identical to those in the article (We used the copied and paste option)

The author list is identical

The name of table 3a is changed into table 3, and table 3b is renamed table 4. (both in the references in the tekst and in the title of the tables.)